

'Making Cadder Better'

ALLOCATIONS POLICY

This document can be made available in different formats, on request, such as Braille, large print or tape. It can also be made available in other languages as appropriate.

Cadder Housing Association Ltd 20 Fara Street Cadder G23 5AE

Date Approved: December 2017

Approved By: Management Committee

Review Date: December 2020

ENGLISH This information is available on request in other languages, in large print, in Braille and on audio format. If you would like this information in one of these formats please contact Cadder HA on **0141 945 3282**

POLISH Niniejsze informacje dostępne są na żądanie w innych wersjach językowych, dużym drukiem, językiem Braille'a oraz w formacie audio. Aby otrzymać powyższe informacje w jednym z wymienionych formatów, proszę skontaktować się z Zespołem ds. Cadder HA pod numerem telefonu **0141 945 3282**

FRENCH Ces informations sont disponibles sur demande dans d'autres langues, en gros caractères, en braille et en format audio. Si vous souhaitez obtenir ces informations dans l'un de ces formats, veuillez contacter Cadder HA au **0141 945 3282**

هذه الدمع لومة م توفرة تحت الطلب بلغات أخرى، بطباعة بأحرف CIBARA ت ترغب في آبيرة، بطري قة برايل وعلى شريط صوتي. إذا آن الدحسول على هذه الدمع لومة بأي من هذه الصيغ، الرجاء أن تحل بفريق سياسة جمعية آلا سكول الإسكان Cadder HA 1945 على الدرقم 2882على الدرقم 1945 على الدرق م

SOMALI Warbixintaan waxaa, haddii la dalbado lagu heli karaa luuqaddo kale, daabacaad weyn, Farta ay dadka indhaha la' akhriyaan (Braille) iyo qaab cajaladdo maqal ah. Haddii aad doonayso inaad warbixintan ku hesho mid ka mid ah qaababkaas, fadlan kala xidhiidh Kooxda Xeerarka ee Cadder HA telefoonka **0141 945 3282**

ف در شت یا israFاین مطالب را می تو انید به زبان های دیگر، به شکل چاپ با حرو حروف بریل (بررای نابیان) و برروی نوار صوتی درخوا ست نمایی ید. در صورتی آه مایل به دریافت این مطالب به یکی از شکل های فوق هستید لطفاً با دفت Cadder HAت ماس تالی به یکی از شکل های فوق هستید لطفاً با دفت شماره آنید حاصل تالی نشماره آنید حاصل

0141 945 3282

RUSSIAN Данная информация может быть предоставлена по требованию на других языках, крупным шрифтом, шрифтом Брайля и в аудиозаписи. Если вы хотите получить данную информацию в одном из этих форматов, обратитесь в Cadder HA по телефону **0141 945 3282**

Cadder Housing Association Ltd

Allocations Policy

Content

Section	Policy Area	Page No
1	Introduction	4 - 8
2	Legal & Good Practice Framework	8 - 16
3	Strategic Aims & Association Values	16
4	The Allocations System	17 - 21
5	Removal & Suspensions	21 - 26
6	Letting Considerations & Requirements	26 - 29
7	Assessing Housing Need	29 - 30
8	The Points System	31 - 42
9	Housing List Review	42
10	Tenancies	43 - 44
11	Other Types of Lets	44 - 50
12	Housing Options & Support	50 - 51
13	North West Common Housing Register	52
14	Complaints & Appeals	52 - 53
15	Audit, Monitoring & Review	53 – 54

Allocations Policy

1.0 INTRODUCTION

1.0.1 Cadder Housing Association

- 1.0.2 Cadder Housing Association is a registered social landlord, set up in 1994 to improve the living conditions for residents in the Cadder estate in the north of Glasgow. It operates on a "not-for-profit" basis, and is run by an elected Management Committee, consisting mainly of local residents who employ a staff team to manage the Association on a day-to-day basis.
- 1.0.3 The Association currently owns and manages 641 properties. These mainly consist of tenemental properties, but also include semi-detached and terraced houses. The Association receives an average of 50 properties for re-let each year, the majority of which are two-bedroom tenemental properties.

1.1 Policy Principles

- 1.1.1 This document outlines the Association's Allocations Policy, which is one of the key documents we produce. In line with legislative and good practice requirements the underlying principles are for it to be: -
 - > Consistent whereby applicants in similar circumstances are considered in a similar manner.
 - ➤ **Open** with not only the policy document being available to potential applicants but also, meaningful information being provided at all stages of the allocations process. Individual confidentiality will be maintained at all times.
 - ➤ **Responsive** the policy document will enable staff to respond to a wide range of needs and circumstances. We will generally allocate properties to applicants with high levels of housing need, which will relate to those needs outlined in the policy.
 - Accountable We aim to provide an allocation service which is not only accountable, but is also transparent in communication with applicants at the different stages of the allocation process and observe confidentiality at all times. We will ensure that all allocations are carefully checked and that the household being

offered a property outlining our decision making processes. Clear audit trails will be maintained which will demonstrate that we have followed all proper procedures in selecting applicants for an offer of housing with due regard to data protection. Audit trails will be made available for inspection by any regulatory body or appeals process.

- Non-Discriminatory the Association's aim is to observe equal opportunity requirements to ensure equality of treatment for all applicants without discrimination or prejudice based on the grounds of gender reassignment, sex, sexual orientation, marriage and civil partnership, pregnancy and maternity, race, ethnic origin, nationality, religion or belief, age, disability, social or financial status.
- Confidential all information provided for the purposes of the assessment for re-housing will be treated as strictly private and confidential under the terms of the Data Protection legislation. It will not be passed onto or discussed with any other person or organisation without the applicant's permission, apart from where the Association is legally required to do so. Where Committee members are required to consider an individual case the identity of the applicant will be treated as confidential.
- Accessible the Association will strive to ensure that this policy is made available on request in a variety of formats, also that applications for housing are available from a number of locations and organisations. We will also ensure that housing in the Cadder area is promoted through marketing initiatives.

1.2 Policy Aims & Objectives

- 1.2.1 The aims and objectives of this policy are to: -
 - Ensure it complies with all relevant legislation and good practice;
 - Ensure that allocation practice is carried out effectively and efficiently to minimise the period properties are empty thus maximising income;
 - Monitor allocation practice through effective performance management systems. Indicators to measure performance will be developed and information will be presented quarterly to the Management Committee and annually to all Cadder residents through the Annual Report and Annual Report on the Charter;
 - Address the housing needs of a range of applicants;
 - Be clear and easy to understand;

- Have a system of prioritising applicants which is fair and consistently applied;
- Allow the Association to manage its rented stock effectively, thereby making best use of the stock;
- Maintain a balanced and stable community;
 - Offering the most appropriate houses to meet people's needs
 - Housing people in appropriate locations
 - Where possible, avoiding clashes of lifestyle by over concentrations of household type or need in one close, street and parts of the Cadder area
 - ◆ Taking account of local need
- Ensure the accommodation available meets the housing need of the applicant;
- Allow Association tenants to transfer to other properties if they have:
 - ♦ Housing need
 - ♦ Change of circumstance
 - ♦ Aspiration to move
- Encourage co-operation between the Association, other housing providers, agencies and Glasgow City Council in their homeless role and other housing providers to meet the needs and aspirations of all applicants;
- Provide good quality housing at an affordable rent;
- Training on the Allocation Policy will be given to staff, Committee members and any operating Resident Group;
- We shall deal with complaints / appeals in relation to any decisions in the implementation of this policy in accordance with our Complaints Policy, as outlined in the Policy;
- ➤ A formal review of this policy will take place every three years, although, operational amendments may be presented to the Management Committee on an annual basis. Where the proposed changes affect service users, they will be consulted prior to the review. The outcome of the review will be published in our quarterly newsletter.
- 1.2.2 We will actively promote access to our list and encourage people to apply for housing in the Cadder area.

- 1.2.3 A summary policy booklet accompanies this policy. This will be provided to all new applicants and where required to existing applicants, so as to give an understanding of how we re-let properties. It will also be on display and available at the Association's reception area.
- 1.2.4 The Allocations Policy will also be available on the Association's website.

1.3 Commitment

- 1.3.1 Our commitment to all applicants who apply for housing is:-
 - To respect confidentiality;
 - ❖ To provide good quality information in plain language;
 - To assess all applicants details on their merits and in accordance with the policy;
 - To record and where necessary request information relating to the application, for example personal details, tenancy references etc.;
 - We will acknowledge receipt of the application for housing within five working days;
 - ❖ To make the applicant aware of their provisional points total, queue position and prospects of securing accommodation in areas of Cadder they have requested within ten working days:
 - We will provide information to the applicant on properties and areas they have not requested but would qualify for, which may result in an earlier offer of housing;
 - Where there is unlikely to be an offer of housing in the Cadder area within a reasonable period of time, we will provide applicants with details of other social rented landlords in the North West area of Glasgow (Many of which form part of the North West Glasgow Housing Register);
 - Staff will be available to give information and advice on applications, the allocations process, prospects for housing and other related matters either at the office, in writing or by telephone.

1.4 Equal Opportunities

1.4.1 All applicants aged 16 years and above will be accepted onto our list.

- 1.4.2 The Association will regularly test this policy and associated procedures for equal opportunity implications and take appropriate action. To ensure equal access to the information contained in this policy for all, we are happy to provide copies in Braille, in larger print, translated into other languages or audio format to you or anybody that you know.
- 1.4.3 The Association will ensure that the Equality Act 2010 is adhered to during its operation of this policy. We will ensure that all residents and other customers are treated fairly across all Association policy and practice areas.

1.5 Consultation

- 1.5.1 Consultation with the community and resident participation has always been one of the key aspects of the way in which the Association operates. We therefore, welcome the statutory provisions of the Housing (Scotland) Act 2001, which require all social landlords to take tenants' views into account when formulating key service policies.
- 1.5.2 There are two strands to policy consultation within Cadder: -
 - Firstly, we believe it is important to inform all tenants when a policy review is due, together with some information on the proposed changes by the Association; and
 - Secondly, to invite feedback from tenants on (i) how well they feel the proposals address the needs of the community and (ii) whether there are any particular amendments they would like to see.
- 1.5.3 The Association's Resident Participation Strategy outlines our commitment to providing information and consulting with residents on our policies and in particular, the ones that affect them.
- 1.5.4 The Association takes account of feedback from tenants, service users and relevant external agencies in the review of policies.

2.0 LEGAL & GOOD PRACTICE FRAMEWORK

2.0.1 This policy takes account of current legislation and good practice guidance. The following is a brief summary of the legislation that affects the allocation of properties:

2.1 Housing (Scotland) Act 2001

2.1.1 The Allocations Policy adheres to the Housing (Scotland) Act 2001. The policy contains guidance on admission to its housing lists, how properties are allocated, allows for the transfer of tenants to other

landlords, and allows for the exchange of houses. The policy is available to the public in its entirety and in summary form.

- 2.1.2 Section 5 of the Act imposes a duty on the Association to house homeless applicants referred to us by the Local Authority, unless we have good reason not to.
- 2.1.3 Section 54 of the Act obliges Housing Associations to consult with tenants, service users and Registered Tenants Organisations regarding significant changes in relevant Association service areas.
- 2.1.4 Section 106 of the Act requires the Association to 'act in a manner which encourages equal opportunities and, in particular, the observance of equal opportunities requirements. Equal opportunities are widely defined as: -

'the prevention, elimination or regulation on discrimination between two persons on the grounds of disability, age, sexual preference, language or social origin or other personal attributes including beliefs or opinions such as religious beliefs and political opinions'.

2.2 Lets to Staff, Former Staff, Board Members, Former Board Members or Staff / Board Family Members

- 2.2.1 It is extremely important that lets to staff, Board or members of their families are transparent and accountable at all times.
- 2.2.2 Where an property allocation (or other payment / benefit) is being made to a member of staff, Board or members of their families (also applies for former Staff and Board Members) a written report from the Customer Services Team Leader / Customer Services Manager will be presented to the Board recommending the allocation and confirming it complies with the Association's Allocation Policy.
- 2.2.3 In order to protect our reputation and demonstrate that we conduct our affairs with openness, honesty and integrity, we maintain a Register of Interests. Staff and Committee must record in this register any interests that they or someone connected to them has, which is relevant to our business. Staff and Committee will be required to confirm annually that their entry is accurate and up to date.
- 2.2.4 A tenancy agreement cannot be signed until Board approval has been given.
- 2.2.5 The Association's Entitlement, Payments & Benefits Policy describes the entitlements, payments or benefits that our people are able to receive. It also describes what is not permitted and the arrangements that we have in place to ensure that the requirements of this policy are observed. This policy also outlines the Association's approach to managing lets to 'connected people'.

Connected people are defined as:

Group 1 Members of your household	Group 2 People closely associated with you	Group 3 Others you need to consider
Anyone who normally lives as part of your household, whether they are related to you or not, including spouses/partners who work away from home and sons and daughters who are studying away from home	 Parents, parents-in-law and their partners Sons and daughters; stepsons and step-daughters and their partners Brothers and sisters and their partners A partner's parent, child, brother or sister Grandparents, grandchildren and their partners Someone who is dependent on you or whom you are dependent on Close friends 	Other relatives (e.g. uncles, aunts, nieces, nephews & their partners) Other friends (e.g. someone you are acquainted with socially, neighbours, business contacts/associates)

- 2.2.6 The connected person(s) cannot be present during the Board discussion about the allocation and can only return to the meeting once the discussion has finished.
- 2.2.7 If an allocation to a connected person is approved, the allocation must be entered into the Entitlements Payments & Benefits Register.

2.3 Matrimonial Homes (Family Protection) (Scotland) Act 1981

- 2.3.1 The Matrimonial Homes (Family Protection) (Scotland) Act 1981 must be adhered to when providing advice to tenants on the statutory occupancy rights to their home. This legislation introduced rights for non-entitled spouses (non-tenants) where they have:-
 - ❖ A right to remain in the family home;
 - ❖ A right to return to the family home if excluded from it, to enter and occupy the family home;

- ❖ A right to pay rent and require a landlord to carry out his duty to maintain the home;
- ❖ The consent of the non-entitled spouse must be sought for any transaction which would affect his/her occupancy rights e.g. termination of tenancy; and
- ❖ A right to a defence against repossession if the tenant fails to do so him/her self.

2.4 Data Protection Act 1998

- 2.4.1 The Association will adhere to the Data Protection Act 1998 in order to ensure that information provided by an applicant or sought by the Association is relevant only to assessing their housing need and is treated in the strictest confidence.
- 2.4.2 Under the provisions of the Data Protection Act 1998 individuals are entitled to request personal data which is held by the Association. The Association may make a charge covering the administrative costs of providing this information.
- 2.4.3 The General Data Protection Regulation (GDPR) will apply from May 2018, significantly enhancing the requirements under the current Data Protection Act.

2.5 Human Rights Act 1998

- 2.5.1 In accordance with articles 7 and 11 of the Human Rights Act 1998, the Allocations Policy ensures that respect for applicants' private and family life is given when assessing housing need. Actions that interfere with these rights may be open to challenge.
- 2.5.2 In accordance with article 8 of the Act, applicants will not be banned from housing lists. However, their applications might be subject to deferment from housing allocation as per the terms of the Housing (Scotland) Act 2001 and the allocation policy. Any decisions to defer a housing allocation will be thoroughly documented.

2.6 Housing (Scotland) Act 2010

- 2.6.1 The stated purpose of the Housing (Scotland) Act 2010 is to improve value for tenants and taxpayers through a modernised system of social housing regulation. It was also introduced to safeguard social housing for the use of future generations of tenants by reforming the Right to Buy (RTB). RTB for social housing tenants in Scotland subsequently ended on 31 July 2016.
- 2.6.2 A summary of the Act's main sections are as follows:-

- Part 1- Established the Scottish Housing Regulator as an independent regulator with the objective of safeguarding and promoting the interests of tenants, prospective tenants, homeless people and others using housing services provided by social landlords.
- Part 2- Requires the Scottish Housing Regulator to keep a register of social landlords and sets out the criteria for registration and the circumstances in which a body may be removed from the register.
- Part 3- Provides for Scottish Ministers to specify the standards and outcomes social landlords must aim to achieve in a Social Housing Charter, and for the Scottish Housing Regulator to set performance improvement targets and assess and report on their performance. It also requires the Scottish Housing Regulator to set out standards of governance and financial viability for registered social landlords.
- Part 4- Provides the Scottish Housing Regulator with powers to carry out inquiries and obtain information from social landlords.
- Part 5 Gives the Scottish Housing Regulator a range of powers to intervene where it has concerns about a social landlord's performance, governance arrangements or financial viability. It also requires the regulator to issue a code of practice explaining how it will use its intervention powers.
- Part 6 Provides for the Scottish Housing Regulator to set accounting requirements for registered social landlords.
- Part 7 Provides the Scottish Housing Regulator with powers to deal with an insolvent registered social landlord.
- Part 8 Deals with the constitution, rule changes, amalgamation and dissolution of registered social landlords.
- Part 9 Sets out controls on the disposal of land by registered social landlords.
- Part 10 Sets out a special procedure for certain disposals by and restructurings of registered social landlords where there is a change of landlord.
- Part 11 Makes special provision, including approval by tenants, for a change of landlord from a local authority landlord.
- Part 12 Makes provision for regulation of charitable registered social landlords.
- Part 13 Makes provision for long leases and heritable securities where a social landlord or a rural housing body is, or is to be, the lessee or the debtor in the security.

Part 14 - Reformed the Right to Buy.

(The Right to Buy was ended throughout Scotland on 31 July 2016 by the Scottish Government).

- Part 15 Amends the Housing (Scotland) Act 2005. The changes relate to local authority maintenance powers, charging orders and repayment charges, the scheme of assistance and enforcement powers.
- Part 16 Includes miscellaneous amendments to existing legislation.
- Part 17 Sets out supplementary and final provisions.

2.7 Equality Act 2010

2.7.1 The Association within the operation of its Allocations Policy will comply with the Equality Act which protects persons from being discriminated against.

The characteristics that are protected by the Equality Act 2010 are:

- age
- disability
- gender identity and gender reassignment
- marriage or civil partnership (in employment only)
- pregnancy and maternity
- race
- religion or belief
- sex
- sexual orientation.
- 2.7.2 Under the Equality Act, if you have one or more of these protected characteristics, it is also against the law to treat you the same as everyone else if this treatment will put you at a disadvantage.
- 2.7.3 Equalities is supported by Section 106 of the Housing (Scotland) 2001 Act, which states that landlords have a duty to encourage equal opportunities and provide services in a way that promotes equality.

2.8 The Scottish Social Housing Charter

2.8.1 Section 31 of the Housing (Scotland) Act 2010 introduced the Scottish Social Housing Charter which sets the standards and outcomes that all Social landlords should aim to achieve when performing their housing activities. The Charter became effective on 1 April 2012. It placed a requirement on the Scottish Housing Regulator to issue guidance

setting out indicators by reference to which it intends to measure the progress of social Landlords towards achieving standards and outcomes:

Outcome 1: Every tenant and other customer has their individual needs recognised, is treated fairly and with respect, and receives fair access to housing and housing services:

Outcome 2: Tenants and other customers find it easy to communicate with their landlord and get the information they need about their landlord, how and why it makes decisions and the services it provides;

Outcome 3: Tenants and other customers find it easy to participate in and influence their landlord's decisions at a level they feel comfortable with.

Outcome 4: Tenants' homes, as a minimum, meet the Scottish Housing Quality Standard (SHQS) when they are allocated; are always clean, tidy and in a good state of repair; and also meet the Energy Efficiency Standard for Social Housing (EESSH) by December 2020.

Outcome 5: Tenants' homes are well maintained, with repairs and improvements carried out when required, and tenants are given reasonable choices about when work is done.

Outcome 6: Tenants and other customers live in well-maintained neighborhoods where they feel safe.

Outcome 7: People looking for housing get information that helps them make informed choices and decisions about the range of housing options available to them;

Outcome 8: Tenants and people on housing lists can review their housing options;

Outcome 9: People at risk of losing their homes get advice on preventing homelessness;

Outcome 10: People looking for housing find it easy to apply for the widest choice of social housing available and get the information they need on how the landlord allocates homes and their prospects of being housed.

Outcome 11: Tenants get the information they need on how to obtain support to remain in their home; and ensure suitable support is available, including services provide directly by the landlord and by other organisations.

Outcome 12: Homeless people get prompt and easy access to help and advice; and provided with suitable, good-quality temporary or emergency accommodation when this is needed; and are offered continuing support to help them get and keep the home they are entitled to.

Outcome 13: Tenants, owners and other customers receive services that provide continually improving value for the rent and other charges they pay (including rental income).

Outcomes 14 & 15: A balance is struck between the level of services provided, the cost of the services, and how far current and prospective tenants and service users can afford them.

Tenants get clear information on how rent and other money is spent, including details of any individual items of expenditure above thresholds agreed between landlords and tenants.

2.9 Housing (Scotland) Act 2014

2.9.1 The Housing (Scotland) Act 2014 (the Act) received royal assent on 1 August 2014. Part 2 concerns social housing allocations and tenancy management. The Policy will be reviewed in line with these provisions, once a commencement date and further details are confirmed by the Scottish Government.

2.10 Regulation Framework (Scottish Housing Regulator)

2.10.1 This policy is particularly aligned to Standards 2, 3 and 5 of the Scottish Housing Regulator's Regulation Framework:

Standard 2 – 'The landlord is open about and accountable for what it does. It understands and takes account of the needs and priorities of its tenants, service users and stakeholders. And its primary focus is the sustainable achievement of these priorities'.

Standard 3 – 'The RSL manages its resources to ensure its financial well-being and economic effectiveness'.

Standard 5 – 'The RSL conducts its affairs with honesty and integrity'

2.11 Homelessness (Scotland) Act 2003

2.11.1 The Act amends existing homelessness legislation, contained in the Housing (Scotland) Act 1987 and the Housing (Scotland) Act 2001. Section 11 of the 2003 Act places a duty on landlords to notify the local authority when raising proceedings for possession of a dwelling house.

2.12 Scottish Government & Chartered Institute of Housing

2.12.1 This policy has been reviewed in line with the Scottish Government document, 'Social Housing Allocations: A Practice Guide' and the 'Suspending Housing Applicants Guide' (CIH / TC Young).

3.0 STRATEGIC AIMS & ASSOCIATION VALUES

3.1 This Policy is aligned and informed by the Association's Strategic Aims for 2011-18 and its Values. These are:

3.1.1 Strategic Aims:

- Provide a high quality housing service that is continually responsive to the expectations of our tenants and other customers;
- II. To engage and build relationships with our customers to ensure our service and activities meet their needs and aspirations of our tenants and other customers;
- III. To invest in our people to ensure they have good knowledge and skills to excel in their role within the Association;
- IV. Pursue development, regeneration and wider role initiatives in close working with key partners with the aim of improving Cadder, as well as the quality of life and living conditions of tenants and residents in Cadder; and
- V. Maintain the financial viability of the Association through sound business planning, control and achievement of best value in all that we do.

3.1.2 Association Values:

The Association identified core values associated with our commitment to improve the 'customer journey' in our services, which was integral to our achievement of Investors in People (Silver) accreditation in July 2015. These values (Our 4 'Cs') are:

- ✓ Customer Focussed
- ✓ Communication
- ✓ Caring

✓ Commitment

4.0 THE ALLOCATIONS SYSTEM

4.1 The Association operates a 'Group Plus Points' system, where applicants will receive points relating to their circumstances and housing need.

4.2 Access Routes / Applicant Status

4.2.1 Reasonable Preference

- 4.2.1.1 The law requires landlords to give reasonable preference to certain groups when letting properties. The groups to which landlords must currently give reasonable preference when letting houses are:
 - (a) homeless people and those threatened with homelessness
 - (b) people living in:
 - housing below the tolerable standard
 - overcrowded houses or large families
 - unsatisfactory housing conditions
- 4.2.1.2 As detailed, the Reasonable Preference Groups will be amended within the Housing (Scotland) Act 2014. The Scottish Government has yet to announce a commencement date.
- 4.2.1.3 It is acknowledged that people may be in housing need for other reasons than those covered in law and such cases will be considered for housing with the Association.

4.3 Housing List Categories

- 4.3.1 The Association's Allocations Policy categorises applicants from their status and access routes into category groups. These are as outlined below.
- 4.3.2 The Association sets annual targets for each category group. The Management Committee is presented with a report each year to approve the letting targets for the forthcoming financial year. This approach aims to ensure a balance of lets from the various access routes. To establish the letting targets the Association analyses information from the following sources:
 - Historical lets by access route / applicant status over recent years
 - Current housing list
 - Allocations Policy

- ♦ Good Practice Guidance
- ◆ Legislative compliance through Protocol Agreements, for example homeless persons
- 4.3.3 The current category groups with the Lettings Targets are attached as Appendix 1.
- 4.4 A summary of each category group is as follows:

4.4.1 Waiting List

4.4.1.1 An applicant on the waiting list will be a person(s) who is not a tenant, joint tenant or is the subject of a referral from either the Local Authority or another Registered Social Landlord.

4.4.2 Transfer List

- 4.4.2.1 Current tenants have the right to apply for re-housing in exactly the same way as waiting list applicants.
- 4.4.2.2 The Association recognises that some tenants have no apparent housing need for a change of house as outlined in the policy but may wish to move to an Association house they aspire to. Applicants on this list will be considered in the order in which the Association receives them. This category will not exceed 5% of the annual number of lets.
- 4.4.2.3 In addition to the criteria relating to outstanding monies and behaviour, tenants being offered tenancy transfers will be subject to a satisfactory property inspection. Where issue(s) are identified, an officer will advise the tenant how matters can be resolved to ensure they can progress their offer of re-housing. If such issues cannot be readily resolved, the tenant will be advised that the available offer of housing is to be withdrawn and the required actions to be taken to allow for future offers of housing to proceed. Where necessary, the tenant's housing application will be suspended until required actions are carried out by the tenant. As with all other cases, the Association will regularly review such suspensions and liaise with the tenant during reviews. Tenants registering for a transfer will be encouraged to contact the Association to arrange a tenancy inspection, which will allow for any issues to be identified at an early stage and also allow for housing prospects / options to be discussed.
- 4.4.2.4 As a result of Central Government's Welfare Reform Act, working aged tenants who are deemed to be under-occupying their homes have, from April 2013 been subject to a reduction in any payable housing benefit. This is commonly known as the 'Bedroom Tax' (Social Size Criteria).
- 4.4.2.5 Affected tenants will experience reductions of 14% for one spare bedroom and 25% for two or more spare bedrooms. The deduction

amount will be taken from the actual gross rent amount as opposed to the paid housing benefit amount. The Social Size Criteria remains to be in place and affected tenants are currently being protected against reductions through awards of Discretionary Housing Payments. These are payments funded by the Scottish Government and administered by Glasgow City Council (Housing Benefit Department). The Social Security (Scotland) Bill 2017 intends, amongst other benefit areas, to abolish the Social Size Criteria throughout Scotland. However, there are concerns as to how the UK Central Government will respond to this and even if successfully achieved, an abolishment date will require to be legally set.

- 4.4.2.6 Priority will continue to be given to such tenants seeking to move to a smaller house as a result of the financial pressures that this situation will cause. It is recognised that a large number of tenants who may wish to transfer to another property to avoid financial hardship as a result of the Social Size Criteria will ideally require 2 apartment properties. However, the availability of such properties is extremely scarce and given this, tenants under-occupying by two bedrooms or more will also be considered for 3 apartment properties so as to improve options and reduce the effect of the Bedroom Tax, which would result in the tenant experiencing a 14% reduction as opposed to a 25% reduction.
- 4.4.2.7 Tenants seeking to transfer to a smaller property as a result of the Bedroom Tax will be considered for equivalent type housing.
- 4.4.2.8 Tenants transferring to smaller properties will not only assist their own needs but also the needs of other applicants' on the Association's Housing List by the freeing up of larger type housing thus making best use of our housing stock. This is in line with the reasonable preference provisions of the Housing (Scotland) Act 2014, which will encourage tenants to down-size, so as to release larger accommodation.

4.4.3 Homeless Case Referrals

4.4.3.1 The Association as a Registered Social Landlord (RSL) has since the inception of the Housing (Scotland) Act 2001 had a statutory duty to provide accommodation for homeless people.

The duty to house homeless persons will be met in the following ways:

- The Association will provide information to the Local Authority in respect to its housing stock, for example size, type and turnover of properties.
- ❖ The Local Authority, Glasgow City Council, will assess homeless persons and referrals will be made under section 5 of the Act. We will consider the referral and reply to the Homeless Caseworker within five working days to confirm our acceptance

or refusal of the referral. If accepted we will confirm a likely timescale for the provision of accommodation. We will assess each referral but will not unreasonably refuse a homeless referral for accommodation. Reasonable grounds for refusal may include: -

- Where re-housing is likely not to be secured within the target timescales due to poor turnover of size or type of property;
- Improvements to property not suitable for that particular homeless case, for example Adapted properties.
- 4.4.3.2 The Association will liaise with the local authority to monitor, review and contribute to homeless allocations and strategies aimed at housing homeless people in the Cadder area.
- 4.4.3.3 We will be pro-active in identifying and referring applicants who may be a 'priority group' under the Homeless person's legislation to Glasgow City Council for a full assessment under the Act. We will complete an application form for housing and this will be processed as a waiting list case up until a formal homeless referral has been received and accepted.

4.4.4 Referrals from Other Agencies

- 4.4.4.1 We will set aside a percentage of our lets each year for referrals from other social landlords, as well as housing and support agencies. A referral will only be considered where the access routes to housing cannot meet an applicant's housing needs and aspirations.
- 4.4.4.2 The Customer Services Team Leader / Manager will assess and consider for approval each referral received by the Association.

4.4.5 Medical Priority Cases

4.4.5.1 The Association makes best use of its housing stock, for example we allocate ground floor all on one level accommodation or adapted properties to cases with medical needs.

4.4.6 Management Transfers

- 4.4.6.1 The Customer Services Team Leader / Customer Services Manager will have discretion to make an allocation of housing to a tenant where the normal housing need criteria would constrain a transfer of home. In these cases a transfer will be seen as essential to deal with an urgent tenancy related matter.
- 4.4.6.2 For example, an anti-social problem due to external violence and/or ensure the safety of the tenant and their family. These cases will be

thoroughly documented and be included in the next reports to the Management Committee for their information. In view of the nature of management transfers, it is anticipated that these will only be used on an exceptional basis.

4.4.7 Ex-Service Personnel

- 4.4.7.1 Although not a category group, in accordance with the Scottish Government's 'Social Housing Allocations Practice Guide', the Association recognises the challenges faced by personnel leaving or being discharged from the armed forces and will through the guidance:-
 - ❖ Award the same level of priority to ex-service personnel as those with a similar level of housing need;
 - Give consideration to injured ex-service personnel who require adapted housing as a result of their injury / disability through our medical assessment process in conjunction with the armed forces medical and rehabilitation services:
 - Not impose residency or local connection criteria which may disadvantage ex-service personnel from fair and equal access to our available housing stock;
 - Will be mindful of and supportive to the needs of ex-service personnel and their widows/widowers/civil partners and conform to section 5.3(b) of the guidance.

4.4.8 Asylum Seekers and Refugees

4.4.8.1 People subject to immigration control must declare this when completing their application. Eligibility for housing will require to be assessed in terms of the Housing (Scotland) Act 2010 and asylum and immigration legislation. People may remain on the housing list, but will be asked to provide evidence of their immigration status before we can confirm if they are eligible for an offer of housing.

5.0 REMOVALS & SUSPENSIONS

5.1 In accordance with section 9 of the Housing (Scotland) Act 2001, the Association will not exclude anyone aged 16 years or over from submitting an application form for housing. As such, we will fully assess and log all received applications.

5.2 Removal

5.2.1 There Association will only cancel an application for housing in the following circumstances:-

- The applicant has requested removal from the housing list in writing;
- The death of the applicant;
- ❖ The failure of the applicant to respond to a periodical review of their application. The Association will not automatically cancel an applicant's application when s/he fails to respond to the first letter issued as part of the annual review process and as such, further effort(s) will be made to contact the applicant after initial non-responses before consideration is given to cancelling applications. This will as a minimum, include a follow-up letter being issued, which will inform applicants of a cut-off date for re-registration.

5.3 Suspension

- 5.3.1 In some applications there will be legislative or policy guidance that necessitates the need to suspend the applicant's application from the Housing List. We will:-
 - Carefully consider and evaluate the evidence in relation to the breach of legislative and/or policy guidance;
 - Subject decisions to a test of reasonableness taking full account of the applicants housing need and circumstances;
 - Give clear information to the applicant on how they can end the suspension of their application from the housing list;
 - Regularly review the suspensions on our housing list;
 - Have clear and accountable appeal mechanisms in place for cancellation and suspension cases;
 - Provide information on suspensions to our Management Committee and the Scottish Housing Regulator through the Annual Return on the Charter (ARC).
- 5.3.2 The Association will carefully consider the period that applications for housing are suspended. We will set a timescale that we feel will allow the applicant to resolve the breach of policy and/or legislative guidance and have their application reinstated into the live category of the housing list. We will make the applicant fully aware of the reason for the suspension and the associated timescales.
- 5.3.3 The Association is committed to providing good quality information to all applicants on their prospects of housing. In the information to applicants that are being suspended from the housing list we will confirm:-

- The reason for the suspension from the housing list
- ❖ Information on how the applicant can end their suspension
- The applicant's right of appeal
- The review period for the suspension

5.4 Rent Arrears & Other Debts

- 5.4.1 Where the applicant is an existing tenant of an RSL or Local Authority, amounts of up to one month's rent outstanding will be disregarded. Where the applicant has more than one month's rent outstanding and is a tenant of a RSL or Local Authority a satisfactory repayment agreement will require to be in place with the landlord which has been maintained for at least three successive months.
- 5.4.2 Where a housing list applicant has monies outstanding amounting to more than one month's rent or from a previous tenancy a satisfactory repayment agreement will be required to be in place with the previous landlord and maintained for at least the three successive months immediately preceding the allocation.
- 5.4.3 We shall not suspend applicants with debts that are five years or older.
- 5.4.4 In the event of an applicant having other outstanding tenancy related debts, for example rechargeable repairs, such an applicant will be suspended from the housing list and will only be allowed to register their housing need if any of the following circumstances apply:-
 - The monies outstanding are greater than one twelfth of the full amount initially due;
 - The tenant has not entered into a repayment arrangement;
 - The repayment arrangement has not been maintained for at least 3 months and the applicant is continuing to make these payments.
- 5.4.5 The Association will not be obliged to award a tenancy to an applicant who has been issued with a Notice of Proceedings for Recovery of Possession under Section 14 of the Housing (Scotland) Act 2001 on the grounds listed in paragraphs 1 7 of Schedule 2, Part1 of that Act. Such an applicant will, however, be allowed to register their housing need.

5.5 Unsatisfactory Tenancy Report

5.5.1 Where the Association is made aware that the present or a previous

tenancy has not been conducted in a satisfactory manner it may consider suspension of an application or withdrawal of an offer of accommodation. This decision will depend on the individual circumstances, and staff will discuss the content of any information or tenancy report with the applicant.

- 5.5.2 Where the tenancy breach has not occurred in the past twelve months, it will not count as a reason for withholding the offer. Otherwise, suspension will be for a period that allows the applicant to resolve the tenancy matter to the satisfaction of the Association or full period of twelve months, whichever occurs first.
- 5.5.3 An example would be where a Notice of Proceedings for Recovery of Possession has been served on grounds 1 7 of Schedule 2, Part 1 of the Housing (Scotland) Act 2001
- 5.5.4 The Association will also take into account the condition of tenants' homes, where it is established that neglect of the property is significant and constitutes a breach of tenancy. Where a suspension is applied in such a circumstance, support and advice will be offered.

5.6 Aggressive & Violent Behavior towards Staff

5.6.1 The Association reserves the right to reject applicants who verbally or physically abuse or threaten staff or its contractors. Similarly any applicant who attempts to bribe a staff member or offer some other form of inducement in order to receive an offer of housing will automatically be suspended from the list. This is supported by the Association's Bribery Policy.

5.7 False or Misleading Information

- 5.7.1 Anyone who knowingly provides false information (or withholds something known to be relevant) will have their application suspended.
- 5.7.2 If, after a let has been made, it comes to light that false information may have been given, the Association will consult its solicitor to assess the likelihood of an action for repossession being successful.

5.8 Offers of Housing

- 5.8.1 In the event that an applicant fails to respond to offers of housing on two occasions, their application will be suspended until they have indicated their interest in securing an offer of housing, as well as their areas of choice.
- 5.8.2 Applicants who refuse three reasonable offers of housing will have their application for housing suspended until they have a prospects interview to establish their aspirations for housing in the Cadder area.

5.9 Review of the Housing List

5.9.1 The Association will carry out an annual review of all applications. This exercise will be done on a monthly basis on or around the anniversary submission dates of applications. This will help to ensure that the Association has up to date information from applicants and preferred choice of locations so as to reduce offers of housing that the applicant may otherwise not be interested in and in turn improve allocations and void processes. Applicants who fail to register their interest to remain on the housing waiting list will have their application cancelled and removed from the list.

5.10 Anti-Social Behaviour

- 5.10.1 The Association will normally refuse a tenancy to an applicant where there is evidence of serious and/or sustained anti-social behaviour by the tenant or a member of the household, which has been confirmed by an official source / investigation.
- 5.10.2 This behaviour is likely to have been related to their tenancy, and/or have caused significant alarm, distress, nuisance or annoyance at or in the vicinity of their home or the Cadder area and will normally (but not always) have occurred on a number of occasions.
- 5.10.3 The Association will follow the Chartered Institute of Housing's guidance on suspensions, which states that landlords should take account of the following when considering suspending an applicant's application for housing:
 - the extent to which the conduct is because of acts or omission of people other than the tenant;
 - the nature, frequency and length of the conduct;
 - the effect the conduct is having on other people; and
 - any other action taken by the landlord to address the conduct.
- 5.10.4 Section 34 of the 2001 (Schedule 6) Act gives social landlords the power to provide a Short Scottish Secure Tenancy (SSST). This includes to persons who landlords have evicted for antisocial behaviour in any part of the UK and from any tenure. Social landlords can also offer it to tenants and household members who are under an antisocial behaviour order (ASBO). In both cases the landlord must give support to tenants to help them sustain the tenancy and convert it to a full Scottish Secure Tenancy (SST). These tenancies convert automatically to a full Scottish Secure Tenancy (SST) after 11 months, if there has been no repetition of antisocial conduct. Landlords can

raise proceedings for recovery of possession of the house under a SSST after serving a Notice on the tenant. If the tenant refuses to leave, the courts must give an order for recovery of possession if correct procedures have been observed.

- 5.10.5 Applicants whose housing applications are suspended on the grounds of anti-social behaviour will be advised of the reason and will have access to the appeals process.
- 5.10.6 Where appropriate, the Association will seek to support the tenant to encourage and attain the required improvement in conduct which has caused the suspension. Support will be made available through Associations staff, its tenancy support partners and other relevant agencies.

5.11 Deferred and Delayed Applications

- 5.11.1 The Association will defer applications when an applicant is not in a position to accept an offer of accommodation. For example, applicants with tied accommodation or applicants who are not able to receive offers of housing for reasons such as ill health, bereavement or other personal circumstances.
- 5.11.2 The Association shall give consideration to the progress of applications where there has been a delay in the applicant providing information. Such delays may not necessarily result in offers of housing being withdrawn and may be used as a prompt for securing requested information.

6.0 LETTING CONSIDERATIONS & REQUIREMENTS

6.1 The Association is committed to letting properties in accordance with the aims and objectives of this policy.

6.2 Disregarded Factors

- 6.2.1 The Association will <u>not</u> take into account any of the following factors in the allocation of houses. These are:
 - ❖ The length of time for which an applicant has resided in its area.
 - Any outstanding liability (such as rent arrears) attributable to a house of which the applicant was not the tenant.
 - Any rent or other arrears accrued by the applicant on a previous tenancy, which are no longer outstanding
 - Any such liability which is outstanding but where:

- ◆ The amount outstanding is not more than 1/12th of the annual rent amount payable
- The applicant has agreed repayment arrangements with the landlord for paying the outstanding liability
- ◆ The Association may refuse to house an applicant if she/he has significant outstanding arrears or service charges
- The age of the applicant provided that the applicant is 16 years of age
- The income of the applicant and his/her family
- ❖ Whether or to what value the applicant or any of the applicant's family owns or has owned heritable or moveable property.

6.3 Applicant Obligations

- 6.3.1 Applicants, who seek and aspire to live in the Cadder area, will be required to fulfil the following obligations:
 - Applicants are required to complete a Housing Application form and answer all questions honestly and to the best of their knowledge
 - Applicants will be required to sign a disclaimer allowing details to be sought in relation to tenancy references. Applicants will be required to confirm any change of circumstances in relation to their housing or personal situation
 - Applicants are required to provide information to support their application for housing, for example, proof of residency etc
 - Applicants will be required to confirm their reasons for refusing an offer of housing

6.4 Proof of Residency

- 6.4.1 All residents will have to provide proof of their personal identity and confirmation that they reside at their current address. In addition, they will have to provide information to verify the current address of each person who is part of the family composition.
- 6.4.2 This information will be required before any tenancy is awarded. However, it will be made clear applicants that they should submit required information as soon as possible and failure to do so may result in their application being suspended until requested information is received. We will endeavour to avoid this.

6.5 References

6.5.1 The Association will only seek information on the 'conduct' of an applicant's current or former tenancies within five years. The Association will discuss with the applicant any conduct issue that is highlighted from the reference request and confirm whether these will have an affect in them securing an offer of housing.

6.6 Home Visits

- 6.6.1 The Association will carry out home visits to a random sample of applicants on the housing list.10% of applicants to the housing list will be identified for home visits. This will allow the Association to validate that the points awarded are consistent with the applicants' circumstances and that we are allocating housing to persons in the greatest need.
- 6.6.2 In addition, the Association will carry out a visit to all transfer cases immediately prior to the offer of alternative accommodation. The purpose of this visit will be to assess their property and ensure that it has been maintained in accordance with their Cadder Scottish Secure Tenancy Agreement and to validate household composition with that on the housing application form. Where there is damage to the house, decoration, our furniture or the fixture or fittings that has been caused wilfully, negligently or accidentally by the tenant or anyone living with them or visiting their home, it may result in the withdrawal of the offer of re-housing.

6.7 Balanced Communities

- 6.7.1 The key objectives of the Allocations Policy in relation to maintaining a balanced community are:
 - ❖ To achieve and contribute to the creation of a balanced community, which will realise a healthy variation of residents across the social and demographic spectrums. The Cadder area would then have residents with differing employment experience and different ages and types of households;
 - To offer choice and aspiration amongst people looking for housing.
- 6.7.2 Our approach to maintaining a balanced and stable community will be to:
 - Offer the most appropriate houses to meet people's needs;

- House people in appropriate locations;
- Where possible, avoiding clashes of lifestyle by over concentrations of household type or need in one close, street and parts of the Cadder area.

6.8 Offers of Housing

- 6.8.1 The Association will initially limit the number of offers that an applicant can receive to three reasonable offers of housing. Analysis of the reason for previous refusals will be taken into account when future offers are considered. For example, wrong type of property, not interested in a particular street. The Association will detail the reason for each refusal.
- 6.8.2 After the third refusal the applicant's application for housing will be suspended until they attend a prospects interview to establish their aspirations for housing in the Cadder area. Thereafter, they will be reinstated on to the live list and eligible for three further offers of housing.
- 6.8.3 We shall confirm offers of housing to applicants in writing. Additionally, we shall also telephone the applicant to advise of the available property. If the applicant is interested in the available property, we will arrange a suitable date and time for viewing. If required, the applicant will be allowed 24 hours to make a decision as to whether to accept the offer.
- 6.8.4 If the offer of housing is accepted, arrangements will be made for the applicant to complete missives and other associated paperwork.
- 6.8.5 We shall ensure that all properties are repaired to a suitable and habitable standard and provide a list of identified void property works to the new tenant.
- 6.8.6 Where identified / agreed, we shall ensure that suitable support services are put in place so that new tenants (and existing) can maintain their tenancy and therefore reduce the risk of homelessness occurring.

7.0 ASSESSING HOUSING NEED

7.1 Housing Need

- 7.1.1 The Association will in the main observe the principles of housing need when allocating our properties, and will at all times:-
 - Ensure that in the main, allocations from whatever source are primarily based on housing need;

- Work in conjunction with other housing providers with the aim of complementing its housing provision and housing those in the greatest housing need;
- Ensure the accommodation available is suitable for the applicant's housing need;
- Consider social and medical factors and the capacity to improve the applicant's quality of life;
- Minimise the length of time properties are vacant in order to allocate to those in housing need as quickly as possible;
- Make best use of available stock;
- Seek to maximise stock; and
- Help to sustain communities
- 7.2 Section 20(1) of the 1987 Housing (Scotland) Act as amended requires the Association to give reasonable preference to persons who:-
 - (i) are occupying houses which do not meet the tolerable standard;
 - (ii) are occupying overcrowded houses;
 - (iii) have large families;
 - (iv) are living under unsatisfactory housing conditions;
 - (v) are homeless persons and persons threatened with homelessness
- 7.3 The factors we take into account in assessing housing need are:
 - Overcrowding / under occupation
 - ♦ Medical
 - Marital / Partnership Breakdown
 - ♦ Harassment / Abuse
 - Homeless
 - Lack of, or sharing amenities
 - ♦ Condition of current property.
 - ♦ Social / Family Support
 - ♦ Economic Factors
 - Welfare Reform Act / 'Bedroom Tax'

8.0 THE POINTS SYSTEM

8.1 The housing need of each category is given a particular number of points in order to gauge the level of housing need each applicant is experiencing:

8.2 Overcrowding / Under-Occupation

- 8.2.1 The Association shall award points where there is overcrowding or under occupation of the property, against the following criteria: -
 - One bedroom for the applicant and / or partner
 - ◆ One bedroom for two children of the same gender (where neither child is aged over 11 years of age)
 - ◆ One bedroom for two children of opposite gender (up to the eldest child reaching his/her 9th birthday)
 - One bedroom for all other persons
- 8.2.2 The level of points that can be awarded are as follows:

Overcrowding Points:

One bedroom short - 30 Points

Each additional bedroom short - 15 Points

Under-Occupation Points:

One bedroom extra - 100 Points

Each additional extra bedroom - 20 Points

8.2.3 A household will not normally allocate accommodation that would cause immediate overcrowding or under occupation. The Customer Services Team Leader / Customer Services Manager will have discretion to authorise an offer to an applicant where overcrowding / under-occupation would result, in exceptional circumstances.

- 8.2.4 Tenants under-occupying properties will be entitled to under-occupation points to help meet their housing need in terms of accommodation size. Priority in this regard will be given to tenants who are subject to the Social Size Criteria (Bedroom Tax). Appropriate points based upon number of unoccupied bedrooms will be awarded to each application. Applicants are required to provide evidence of the number and ages of their household. Applicants, other than Association tenants, will not qualify for the initial award of 100 points, and will otherwise be awarded 20 points for each under-occupied bedroom.
- 8.2.5 Applicants must have part custody of, or regular overnight access to their children before they will be considered as part of the household. The Association will require a written declaration from the applicant or another source of information confirming the current access or custodial arrangements of their children.
- 8.2.6 In addition to any other persons included in the applicant's household, the house size allocated to applicants with access only will depend on the number of children and the frequency of visits and duration of residence.
- 8.2.7 Where applicants are being offered a property with an extra bedroom for access purposes we will make them aware that they will or may become subject to the Bedroom Tax so that they may make an informed decision. We will provide advice in relation to payment of rent and offer assistance from our welfare rights service.
- 8.2.8 Applicants requesting an additional bedroom on medical grounds require to complete a medical self-assessment form or alternatively provide information from their G.P or Consultant. If confirmed, the number of bedrooms required will be assessed against the family composition and overcrowding points calculated. Such cases may be subject to the Social Size Criteria (Bedroom Tax) and accordingly clear advice and information will be provided in this regard.
- 8.2.9 Applicants who apply for housing on the grounds of marital or relationship breakdown, but who have to remain within the family home may be eligible for overcrowding points. These points will be awarded unless the property is currently under occupied and the applicant has their own bedroom.
- 8.2.10 In the event of a birth of a child, the Association shall acknowledge the child as part of the family composition on the provision of confirmation on the date of confinement.

8.3 Medical

- 8.3.1 Medical points will be awarded where an applicant or member of their household / application for re-housing requires re-housing on medical grounds due to unsuitability of the current accommodation. A self-assessment medical application form requires to be completed, which will request details of the applicants:
 - Medical condition
 - ♦ Current accommodation
 - ♦ Accommodation that would be suitable for their medical needs
- 8.3.2 The points will be awarded based on the applicant's present accommodation by its design, condition or location, which adversely affects the medical or mobility condition. The level of priority and points will be based on the unsuitability of the property in relation to the applicant or household member's medical condition. In addition, we will take into account the type of property that would be suitable for the applicant's needs in our award of medical points. The following outlines our medical priority awards:

8.3.3 Priority 'A'

- 8.3.3.1 Applicants in this category will normally have severe difficulty accessing and leaving their home and re-housing is extremely urgent.
- 8.3.3.2 Applicants who are given a medical 'A' priority will only be considered for ground floor all on one level accommodation.
- 8.3.3.3 Applicants awarded a medical priority A will be placed on a medical list and be given priority for suitable ground floor or adapted properties, which are suitable for their medical needs. Their list position will be based on their total housing need points awarded.

Medical 'A' Priority: 100 Points

8.3.4 Priority 'B'

8.3.4.1 Applicants in this category will be awarded points where they have great difficulty in accessing or leaving their home or alternatively it is causing serious aggravation to their medical condition. For example, this category could be awarded to assist applicants to secure accommodation, which has considerably less access problems to their front door.

Medical 'B' Priority: 40 Points

8.3.5 Priority 'C'

8.3.5.1 Applicants in this category will be awarded points where their accommodation is causing significant decree of discomfort to the applicant's medical condition. For example, this category could be awarded where applicants require re-housing for improved quality of life or to accommodation, which has less access problems to their front door.

Priority 'C': 10 Points

8.3.6 Priority 'D'

8.3.6.1 Applicants in this category reside in accommodation that is seen as suitable for their medical needs.

Priority 'D': 0 Points

- 8.3.7 The Association may also seek medical advice in certain cases to clarify or seek further information on the content of the self-assessment application form, with the aim to award the correct medical priority award. Only one award of medical priority will be awarded for each application for housing, which will be the highest priority category.
- 8.3.8 In particular cases, the tenant may be able to continue to reside at their present accommodation if medical adaptations are installed to assist them or members of their household. If medical adaptations are an option, the tenant should be encouraged to contact the Occupational Therapist at Social Work Services to have an assessment of their needs undertaken.
- 8.3.9 In all cases where an applicant has an award of a Medical A or B priority and has been selected for a property, the Association will

contact the Occupational Therapist to arrange for an inspection of the property on offer to the applicant to ensure it is suitable for their needs.

8.4 Marital / Partnership Breakdown

- 8.4.1 The Association's policy is subject to the requirements of the Matrimonial Homes (Family Protection) (Scotland) Act 1981. In addition, the Association will recognise same sex partners, cohabitation and civil partnerships.
- 8.4.2 When an applicant applies for housing on the grounds of relationship breakdown, the Association will react sympathetically and advise applicants of their rights under the above Act. Applicants will also be advised to seek legal advice.
- 8.4.3 The Association has stock in a limited geographical area, so it may not be appropriate for the Association to re-house the applicant. Advice and assistance on alternative housing opportunities will be offered.
- 8.4.4 Applicants whose relationship has broken down at the time of the application and whose application is based on this occurrence must complete a declaration confirming their relationship has ended and they are to separate as a consequence.
- 8.4.5 Applicants, under this category, will placed on the waiting list and pointed according to this criteria and their present housing situation. Any award of Marital / Partnership Breakdown points will be reviewed after six months. Choice of areas will be reviewed with the applicant and points awarded in this category may be removed if the applicant's choices are restricted / unreasonable.
- 8.4.6 As outlined, applicants in this category will be eligible for overcrowding points where they do not have their own bedroom in the marital home.

Marital / Partnership Breakdown: 15 Points

8.5 Harassment / Abuse

8.5.1 The Association will consider applications from applicants who are victims of harassment. Harassment and abuse of any kind is extremely serious and can have a detrimental effect on the quality of life and at worst, be life threatening. Claims of this nature will always be taken seriously with safety being paramount. In dealing with such cases, the Association will adopt the following:

8.5.2 <u>Unable to Continue to reside</u>

- 8.5.2.1 Where the applicant confirms that they are unable to continue to reside in their home due to harassment, emergency re-housing will be sought. The Association will seek the assistance of the Homeless Caseworker Team of the Local Authority, Glasgow City Council to provide temporary accommodation. They will undertake a full assessment of the applicant's situation under the provisions of the Homeless Persons legislation, with a view of providing alternative re-housing.
- 8.5.2.2 The Association will assist their assessment of the applicant's situation where they have previously resided in the Cadder area. They will forward a report to the Caseworker within 47 hours outlining their knowledge of the situation and confirming any further investigations. These further investigations could include a full investigation into the alleged harassment and abuse and request for police or social services reports.

8.5.3 Able to Continue to Reside

- 8.5.3.1 In the event that the applicant can continue to reside in their home or can reside temporarily with friends or family and they are clearly suffering from harassment, the Association will consult them on their prospects of re-housing, which may include: -
 - Move within the area;
 - Where an applicant who is experiencing harassment or abuse lives in the Cadder area and the situation could be resolved with a move within the area they will be considered for a Management Transfer;
 - Move out with the area

Where an applicant who is experiencing harassment or abuse lives in the Cadder area and the situation can only be resolved with a move outwith the area, the Association will pursue a referral to another RSL.

8.5.4 Applicants from Out with Cadder

- 8.5.4.1 Where an applicant, who is a tenant of another RSL makes a direct Waiting List application to the Association and claims to be experiencing harassment or abuse, we will encourage the applicant to make contact with their own landlord. They should be advised that if established their landlord could refer their case for consideration as a Housing Association referral.
- 8.5.4.2 Where an applicant is not a tenant and claims to be experiencing harassment, they will be required to provide information to support their

claims of harassment. The safety of an applicant is paramount and in all cases the applicant will be advised and encouraged to contact agencies who can give assistance, for example Women's Aid, Racial Equality Councils, Police Scotland, Social Work Services and relevant housing advice agencies.

- 8.5.4.3 Confidentiality will be observed and is crucial to ensure safety. The Association will not contact and/or provide information to other agencies or individuals without permission. We will never contact the alleged perpetrators which will include other members of the household where the claims are of domestic violence or abuse.
- 8.5.4.4 We will deal with all claims of harassment and abuse sensitively and provide support and assistance to the person.

Harassment / Abuse: 75 Points

8.5.5 Homeless / Insecurity of Tenure

8.5.5.1 Homeless

Applicants who are assessed by the Local Authority, Glasgow City Council, and in respect of whom a referral under Section 5 of the Housing (Scotland) Act 2001 and is accepted by the Association, will be awarded homeless points only and placed on the Homeless List queue for re-housing. The Homeless List will queue applications in date order, which will reflect the date their application was accepted for re-housing under the Act.

Statutory Homeless Referral: 100 Points

8.5.5.2 Applicants who regard themselves as homeless but have no priority need under the Homeless persons legislation will be entitled to Insecurity of Tenure points.

8.5.6 Potentially Homeless

8.5.6.1 The Association should receive written declaration from the applicant or documentation to confirm that the applicant's status is potentially homeless, for example letter from landlord, letter from their mortgage

company, etc. Applicants that are potentially homeless will be awarded basic overcrowding, lacking amenities and insecurity of tenure points. The awarding of such points is to enhance applicants' prospects of obtaining a tenancy and reduce the likelihood of actual homelessness occurring. All applicants in this position will be encouraged to seek advice and support on their options for housing from Glasgow City Council's Homeless Persons Casework Team.

- 8.5.6.2 The Association's commitment to assisting homelessness is evidenced though its affiliation to Glasgow City Council's Homelessness Duty Protocol and membership of the Council's North-West Local Letting Committee, which was formed to provide a better strategic approach to homelessness.
- 8.5.7 Insecurity of Tenure
- 8.5.7.1 Applicants will be awarded insecurity of tenure points in the following circumstances:
 - 1. Where an applicant is in tied accommodation which they are required to leave. Points will only be awarded where the applicant has through no fault of their own had to vacate tied accommodation, for example redundancy, retirement, medical retirement, deletion of job / post etc. No points will be awarded where the applicant has to vacate tied accommodation through his / her conduct, for example dismissal on conduct grounds.
 - 2. Applicants living in mobile homes, boats, caravans or other movable structures that have no place where they are legally entitled to live in them.
 - 3. Applicants who are subject to notice to quit from the owner or tenant of the accommodation. Applicants will not be eligible for points where they have made themselves intentionally homeless, because of, for example, rent arrears or anti-social behaviour.
 - 4. Applicants who have a short Assured tenancy with a private landlord.
 - 5. It has been determined that repossession of the applicant's tenancy or owner-occupied property is inevitable.
 - 6. Applicants living in supported accommodation, for example Richmond Fellowship, Simon Community etc.
 - 7. Where an applicant's home will be demolished by their landlord or repossessed through no fault of their own.

Insecurity of Tenure: 20 Points (One award per application)

- 8.5.8 Lacking / Sharing Amenities
- 8.5.8.1 The Association shall award points to applicants who lack or share the following amenities: -

Lacking Amenities:

No bath or shower: 10 Points

No hot running water: 10 Points

No inside WC: 10 Points

No kitchen facilities: 10 Points

- 8.5.8.2 The maximum number of points awarded in the Lacking Amenities category is 30.
- 8.5.8.3 Sharing Amenities
- 8.5.8.4 Where an applicant is sharing amenities with another household, the following points will be awarded:

Sharing Amenities:

Sharing a bathroom or toilet: 7 Points

Sharing a kitchen: 7 Points

Sharing a living room: 7 Points

The maximum amount of points awarded for sharing amenities is 15.

8.5.9 Dampness / Inadequate Heating

- 8.5.9.1 The Association recognises that all citizens should have a right to live in quality housing accommodation. Applicants will be awarded points where they are living in the following conditions:
 - Rising & penetration dampness
 - Inadequate heating
- 8.5.9.2 The Association will require confirmation of the above before points will be awarded. This will involve asking the applicant to provide satisfactory evidence from the Local Authority's Building Control Department, the Environmental Health Department or their Landlord.

Rising & Penetration dampness: 20 Points

Inadequate heating: 10 Points

8.5.10 Social / Family Support

- 8.5.10.1 The Association will consider the award of social points to applicants who require re-housing because of a serious social problem. Social points will normally only be considered where other solutions to the social issue(s) cannot be resolved through the award of housing need. The award of social points is not designed to be long-term. They are awarded to help secure an offer of housing in a reasonable period of time to help address an ongoing issue, where an award of points is not available from those available within the policy. The award of social points are made on the basis that the situation is of an urgent nature and that the applicant is unable to continue to live in the current accommodation on a long-term basis and as such, alternative accommodation is required within as short a period of time as possible.
- 8.5.10.2 Social points will normally be awarded in the following situations:
 - Applicants who wish to live closer to a relative, friend or any other person in the community to give or receive support;
 - Re-housing of an applicant is required to solve a social problem made significantly worse by their current living conditions.

- 8.5.10.3 Applicants will be required to complete an application for social priority points to detail their circumstances for assessment. Applicants will be entitled to submit supporting evidence with the application.
- 8.5.10.4 Social points will be awarded on the basis that the applicant makes reasonable choices in terms of the location and type of accommodation that s/he will consider. Where the applicant makes limited / restricted choices, which would likely result in an offer of housing not being secured within a reasonable time period, social points may not be awarded.
- 8.5.10.5 We will ensure that detailed discussion is held with the applicant in relation to choices of location and floor levels. Where an award of social points is made, one offer of housing will normally be made to the applicant. Where a reasonable offer of housing is made and is refused by the applicant, the award of social points will normally be removed, unless the applicant can demonstrate why the offer of housing is not suitable. Any decision to make a second offer of housing with social points being retained, will be made by the Customer Services Team Leader / Customer Services Manager.

Social Priority: 20 Points

8.5.11 Economic Factors

- 8.5.11.1 The Association recognises that the allocation of houses can assist applicants to take up employment opportunities or maintain their existing employment.
- 8.5.11.2 To be responsive to these economic factors, applicants who have job opportunities or have a job in the immediate vicinity of the Cadder area and have a travelling journey of in excess of 106 miles or one hour (whichever is the lesser) will be eligible for points. The Association will require written confirmation of an offer of employment or details from the applicant's current employer.
- 8.5.11.3 The Association will be sympathetic in assessing applications for points in this category out with the above guidelines where the applicant works unsociable hours.

Economic Factors: 10 Points

9.0 HOUSING LIST REVIEW

- 9.1 As a minimum, awarded points will be subject to review at annual registration periods throughout the year or at offer of housing stage. Where it is discovered that an applicant's award of points is no longer accurate, applications will be re-pointed and revised points levels will be confirmed in writing to the applicant. Where it is discovered that an award of points is no longer merited at offer of housing stage, the application will be re-pointed and a decision will be made if the offer of housing should remain in place. We will take into account the revised points level, housing need, property location / demand and any other relevant factors.
- 9.2 Applicants will be asked to complete a choice pro-forma at annual reviews, in which the applicant will confirm street choice, house type and acceptable floor levels. This approach will help to mitigate against unwanted offers of housing being made and help to minimise the Association's refusal rate, which is required to be reported annually to the Scottish Housing Regulator.
- 9.3 All offers of housing and any refusal reasons will be recorded on the applicant's housing application file.
- 9.4 The Association will seek to make appropriate offers of housing to applicants at all times and provide best advice as to how applicants can maximise their prospects of obtaining housing with the Association. Where an applicant refuses three offers of housing, a full review meeting will be called with the applicant, so as to mitigate against any future unwanted offers of housing. We will not wait to have discussion with applicants until a third offer of housing has been refused. Staff will have earlier discussions with applicants when any offers of housing are refused.
- 9.5 An initial reminder letter will be issued, in which the applicant will be given two weeks to return the annual registration form. The Association will monitor the non-return of annual review forms and also pursue these by e-mails and phone calls. However, where these efforts do not result in the applicant returning the annual registration form, the application will be cancelled. This will not prevent the applicant from applying again at a future date.

10.0 TENANCIES

10.0.1 The Housing (Scotland) 2001 created a new statutory tenancy and has replaced the previous secure and assured tenancies of the local authority and housing associations.

10.1 Scottish Secure Tenancy

10.1.1 The Scottish Secure Tenancy will be the main tenancy agreement that Cadder Housing Association will use to outline the statutory and contractual rights of responsibility of the Association (landlord) and the tenant.

10.2 Short Scottish Secure Tenancy

- 10.2.1 The Housing (Scotland) Act 2001 introduced the Short Scottish Secure Tenancy (SSST) and Schedule 6 of this Act sets out circumstances in which the Association can offer a SSST. These are: -
 - ❖ Lets to persons previously subject to a repossession order for anti-social behaviour in respect of a tenancy in Scotland, England, Wales or Northern Ireland within a period of three years prior to the service of a notice that a SSST will be offered;
 - Lets to persons where they or members of their household are subject to an Anti-social Behaviour Order granted under section 19 of the Crime and Disorder Act 1998;
 - Temporary lets to persons moving into the area in order to take up employment;
 - ❖ Temporary lets pending development affecting the house;
 - Temporary lets to homeless persons for tenancies of 6 months or over;
 - Temporary lets to persons requiring or receiving housing support services; and
 - Lets in houses leased by the Association from another body where the terms of the lease preclude the Association subletting under a full Scottish Secure Tenancy.
- 10.2.2 The terms of a short SST are for a period of at least six months.
- 10.2.3 Where a short SST has been granted as a result of anti-social behaviour, and there have been no other relevant issues, the short SST will convert to a full Scottish Secure Tenancy after twelve months. The Association will be obliged to support the tenancy in achieving full Scottish Secure Tenancy status by providing housing support services.

- 10.2.4 The Association can seek an order for recovery of possession of the house from the court where we give two months' notice before the end of the lease and we use the recovery procedures available on a full SST.
- 10.2.5 Tenants have a right to appeal to the Sheriff Court where they are aggrieved at being given a short SST.
- 10.2.6 The short SST tenancy rights are identical to the full SST except:
 - ♦ There is no provision for succession
 - Security of tenure is limited

11.0 OTHER TYPES OF LETS

11.1 Succession to Tenancy

11.1.1 In the event of the death of the tenant, the Association will consider applications from occupants of the household to succeed to the tenancy.

They will be considered on the basis of the following levels of priority: -

Level One:

 The tenant's husband or wife or co-habitee if the house was their only or principal home at the time of death;

OR

 a joint tenant, if the house was his or her only or principal home for six months at the time of the tenant's death.

In the case of a co-habitee, she/he must have occupied the house as his/her only or principal home for at least 6 months immediately before the tenant's death.

Level Two:

If no-one qualifies at Level One, or a qualified person does not want the tenancy, it may be inherited by a member of the tenant's family as long as:

- he or she is aged at least 16 at the date of death;
- the house was his or her only or principal home at the date of death.

Level Three:

If no-one qualifies at Level One or Level Two, or a qualified person does not want the tenancy, it will be inherited by a carer as long as:

- he or she is aged at least 16 at the date of death;
- the house was his or her only or principal home at the date of death;
- he or she gave up another only or principal home before the death of the tenant
- He or she is providing, or has provided, care for the tenant or a member of the tenant's family.

In all levels if more than one person qualifies for the tenancy, they must decide among themselves who should get the tenancy. If they cannot agree within 4 weeks of either:

- a. The death of the tenant or;
- b. The date on which the person in question is notified of their right to succeed to the tenancy;

We will decide who will succeed to the tenancy.

- 11.1.2 Where the house has been designed or substantially adapted for a person with special needs, if at the first succession someone qualifies for the tenancy at Level Two or Three and does not have special needs requiring accommodation of that kind, that person will not have the right to succeed to the house but does have the right to be provided with suitable alternative accommodation by the Association. Where, in the case of such a property, at the point of a second succession someone qualifies at Level One, Level Two or Level Three and does not have special needs requiring accommodation of that kind, we have the right to terminate the tenancy and offer that person suitable alternative accommodation
- 11.1.3 If someone qualifies for the tenancy but does not want it, they should tell us in writing within four weeks of the death and leave the house within three months. Rent will be charged only for the actual period of occupation.
- 11.1.4 The tenancy can only be inherited twice under the provisions of the Scottish Secure Tenancy Agreement. If the tenancy has already been inherited twice, the third death will normally end the tenancy. This will not happen if there is a surviving joint tenant whose Scottish Secure Tenancy will continue. However, if there is still a person in the house who would otherwise qualify to inherit the tenancy under the above paragraphs, the tenancy will continue for a maximum of six months

after the last death. The tenancy will not be either a Scottish Secure Tenancy or a Short Scottish Secure Tenancy for that period.

11.2 Mutual Exchange

- 11.2.1 Section 33 of the Housing (Scotland) Act 2001 gives a tenant the right to exchange their house with another tenant, providing that both applicants are tenants and have a Scottish Secure Tenancy Agreement with a RSL or local authority and the tenant(s) have received consent from their landlord(s).
- 11.2.2 Tenants who wish to mutually exchange their home must complete a Mutual Exchange Application Form. Both parties are required to complete this form prior to the exchange being considered.
- 11.2.3 Tenants who wish to transfer with a tenant of a Fully Mutual Cooperative must be agreeable to become a member of the housing cooperative when the exchange takes effect.
- 11.2.4 Tenants must terminate their existing tenancy prior to signing the Scottish Secure Tenancy with their new landlord.
- 11.2.5 We will not unreasonably refuse our consent for a mutual exchange application, reasonable grounds for refusal are as listed:-
 - ❖ A Notice of Proceedings for Recovery of Possession has been served on the tenant under Section 14 of the Housing (Scotland) Act 2001 for grounds listed in paragraphs 1 to 7 of that Act;
 - ❖ An order for recovery of possession of the house has been made against the tenant under section 16 (2) of the Housing (Scotland) Act 2001;
 - ❖ The house was provided by the Association in connection with the tenant's employment;
 - ❖ The house has been designed or adapted for occupation by a person whose special needs require accommodation of the kind provided by the house and if the exchange took place, there would no longer be a person with such needs occupying the house;
 - The accommodation in the other house:
 - is substantially larger required by the household
 - is not suitable to the needs of the tenant or the tenant's family

- ❖ The exchange would lead to overcrowding as defined in section 139 of the Housing (Scotland) Act 1987.
- 11.2.6 Association tenants can register for a mutual exchange through 'HomeSwapper' (www.homeswapper.co.uk) or 'House Exchange' (www.houseexchange.org.uk). These are national mutual exchange data bases and are free of charge to our tenants.

11.3 Assignations

- 11.3.1 The tenant can assign their tenancy to any person that has occupied their home as their own or principal home for at least six months before the date of the written request. The written request should be in the form of a completed Assignation Application Form.
- 11.3.2 We will not unreasonably refuse permission to assign the tenancy, reasonable grounds for refusing permission include:
 - we have served a notice on the tenant warning that we may seek eviction on certain grounds because of their conduct;
 - * we have obtained an order for eviction:
 - the proposed change would lead to the criminal offence of overcrowding; and
 - we intend to carry out work on the house (or the building of which the house forms part) which would affect the part of the house connected with the proposed change.

11.4 Joint Tenancies

- 11.4.1 A tenant can apply for a joint tenancy with any person that has occupied or will occupy the tenancy as their own or principal home. The written request should be in the form of a completed Joint Tenancy Application Form.
- 11.4.2 We will not unreasonably refuse permission for an application for joint tenancy, reasonable grounds for refusing permission include:
 - we have served a notice on the tenant warning that we may seek eviction on certain grounds because of their conduct;
 - we have obtained an order for eviction;
 - the proposed change would lead to the criminal offence of overcrowding;

- ❖ We intend to carry out work on the house (or the building of which the house forms part) which would affect the part of the house connected with the proposed change;
- ❖ The prospective joint tenant is ineligible for a tenancy where they fail to comply with the conditions as outlined in section 4.6 of this policy Cancellation and Suspension of Applicants;
- The prospective joint tenant has been:
 - served with an ASBO
 - evicted for anti-social behaviour in the last three years

11.5 Sub-Letting

- 11.5.1 If a tenant wants to sub-let part or all of the house they must complete a Sub-Let Application Form. They must provide the following information:-
 - Who they want to sub-let your home to;
 - The amount of rent and any other payments (including a deposit) they propose to charge (if any) in the case of subletting;
 - When they want the sub-let to take place;
 - ❖ The tenancy/occupancy terms on which they intend to sub-let (prior to granting consent, we will require sub-lessees are provided with a written agreement and that the terms of the Agreement are acceptable to the Association).

11.6 Lodgers

- 11.6.1 If a tenant wants to take in lodgers, a Permission to Reside Form must be completed. You must provide the following information:
 - Who they want to take in as lodger;
 - The amount of rent or other payments (including deposit) they propose to charge (if any) in the case of taking in a lodger;
 - When the lodging will take place.

11.7 Permission to Reside

11.7.1 The Association requires that anyone wishing to take up residence with an Association tenant, who is not a member of the existing household must apply for permission to reside.

- 11.7.2 The Association will process the application within one month of the written application. If we do not reply within one month, we are deemed to have agreed to your request. If we refuse this kind of permission, we must notify the tenant of the reasons for our refusal in writing within one month of receipt of the application.
- 11.7.3 The Association will not unreasonably refuse an application for permission to reside, reasonable grounds for refusal include, but are not limited to:-
 - There is overcrowding by more than one room;
 - ❖ The applicant has been evicted by the Association in the period 12 months prior to the application;
 - Where the applicant has been served with an interim or full Anti-Social Behaviour Order (ASBO).
- 11.7.4 Where an applicant is refused permission to reside and they apply for housing with the Association, their housing need will be assessed as if they are living at the property. The Association will pursue the tenant to ensure these non-household members remove themselves from the tenancy.
- 11.8 The Association shall process applications for:
 - Succession to Tenancy
 - Mutual Exchange
 - ♦ Assignation of Tenancy
 - ♦ Joint Tenancy
 - ♦ Sub Lets
 - ♦ Lodgers

within one month of receipt of the written application. If we do not reply within one month, we are deemed to have agreed to your request. If we refuse this kind of permission, we must notify the tenant of the reasons for our refusal in writing within one month of receipt of the application.

- 11.9 The examples listed as reasonable grounds for refusal in the above applications are not exhaustive and do not in any way alter our general right to refuse permission on reasonable grounds.
- 11.10 Where applicable, the tenant and any outgoing spouse / partner must complete a Tenancy Termination form to end their interest in the tenancy before the applicant will be able to sign the Cadder Scottish Secure Tenancy Agreement.

11.11 If the tenant / applicant is unhappy about our refusal they have the right to appeal through the Association's complaints procedure or alternatively make an appeal application to the sheriff.

12.0 HOUSING OPTIONS & SUPPORT

- 12.1 Where tenants and other customers approach the Association for accommodation, we will discuss all possible options. This may involve discussion on personal circumstances, housing need, support issues and financial issues.
- 12.2 We will ascertain applicants' situations and provide realistic advice to try help meet their needs.
- 12.3 Discussion with applicants in relation to their housing situation may not only be about the likelihood of being re-housed by the Association, but also on:
 - Private sector letting;
 - Other housing associations
 - Incentive schemes for first time buyers;
 - Shared equity;
 - Supported accommodation;
 - Mortgage to Rent Scheme
 - Adaptations to help tenants stay in their home; and
 - ◆ Downsizing.

12.4 Support Services

- 12.4.1 The Association has a number of support services for tenants and other Cadder residents. These include:
 - Welfare Rights Service
 - ◆ Digital Support Service (IT)
 - ♦ Energy Advice Service
 - Money / Debt Advice Service
 - Impartial Tenancy / Family Support Service (Simon Community & Quarriers)
 - Starter Packs (household items)
 - Antisocial Behaviour Support
 - Handyman Service
 - ♦ Estate Caretaking Service
 - ♦ Interpretation / Translation Services

Partnerships with other organisations:

- ♦ Trussell Trust
- ♦ Glasgow Care Foundation
- Police Scotland

- ♦ Community Safety Glasgow
- ◆ Department of Work & Pensions
- ♦ Social Work Services
- Glasgow City Health & Social Care Partnership
- 12.4.2 The Association extensively promotes its support services through staff, letters, website, reception area and newsletters.

12.5 Association Tenancies

- 12.5.1 Where an Association tenant or other customer is seeking accommodation from the Association, we will give honest and best advice in relation to their situation and maximise their prospects. This will include locations and house types.
- 12.5.2 All new tenants will be advised of the Association's support services. Information on these services will be included in the new tenant's sign-up pack. We will aim to have a Welfare Rights Officer in attendance at all new tenant sign-ups or arrange an appropriate appointment. This will allow for a benefit health check to be carried out, so as to maximize incomes and help the new tenant sustain the tenancy through benefit uptake and applications to the Scottish Welfare Fund and Glasgow Care Foundation. It is of course recognised that not all tenants will require benefits advice or related assistance.
- 12.5.3 All new tenants will receive a settling-in visit within the first six weeks of the tenancy commencing. These visits are important as the attending staff member can provide any required advice or support and also gauge if the new tenant requires help.
- 12.5.4 All homeless referrals will be invited to attend a pre-allocation meeting with the Association. This will focus on the applicant's Resettlement Plan, which is compiled by the applicant's Homelessness Caseworker and outlines any issues / support needs. The objective of this meeting is to be forward-looking and prevent issues occurring so as to sustain the tenancy and prevent the applicant becoming homeless again.
- 12.5.5 Where existing tenants are experiencing tenancy issues, such as non-payment of rent or being the cause of anti-social behaviour, we will offer support to help address issues and sustain the tenancy. We will also be clear to the tenant of the possible consequences should matters not improve.
- 12.5.6 The Association aims to carry out planned home visits to all tenants.

 These visits are aimed at engendering better relationships and identifying any issues that the Association can assist with. This will also help to sustain tenancies.

13.0 NORTH-WEST COMMON HOUSING REGISTER

- 13.1 The Association is a member of the North West Common Housing Register (CHR). A CHR allows persons seeking social housing to apply to any number of landlords within the CHR through a single application form, as opposed to being required to submit an individual application form to each chosen landlord.
- 13.2 Applications received via the CHR will be assessed and pointed in accordance with the Association's Allocations Policy.
- 13.3 The partners of the North West CHR are:

Cernach HA; Charing Cross HA; Drumchapel Coop; Glasgow West HA; Kendoon HA; Kingsridge Cleddans HA; Maryhill HA; ng Homes; Partick HA; Pineview HA; Sanctuary HA and Yorkhill HA.

14.0 COMPLAINTS & APPEALS

14.1 Complaints

- 14.1.1 We will always try our best to give customers an excellent service, but we know sometimes things go wrong and customers might want to complain or appeal against decisions that we make about their housing applications.
- 14.1.2 If you do not agree with the points that we have awarded to your application, you should firstly contact the staff member who assessed your application to discuss matters. This will then be assessed at stage 1 of the Association's complaints' procedure. Stage 2 of our complaints procedure will attend to complaints that require further investigation on issues that customers continue to be unhappy with after completion of stage 1. We will investigate stage 1 and stage 2 complaints within 5 and 20 working days respectively.
- 14.1.3 Not all investigations will be able to be completed within 20 working days. For example, some complaints may be so complex that they require careful consideration and detailed investigation beyond the 20 day limit. However, these would be the exception and we will always try to deliver a final response to a complaint within 20 working days. We will notify customers if we require more than 20 working days to complete our investigations.
- 14.1.4 Once the investigation stage has been completed, the customer has the right to approach the Scottish Public Services Ombudsman (SPSO) if they remain dissatisfied.

14.1.5 The SPSO considers complaints from people who remain dissatisfied at the conclusion of our complaints procedure. The SPSO looks at issues such as service failures and maladministration (administrative fault), as well as the way we have handled the complaint. The SPSO does not normally consider complaints about factoring services, unless they relate to the way we have handled the complaint.

14.1.6 SPSO Details:-

In person: SPSO

4 Melville Street

Edinburgh EH3 7NS

By post: SPSO

Freepost EH641

Edinburgh EH3 0BR

Freephone: 0800 377 7330

Online contact: www.spso.org.uk/contact-us

Website: www.spso.org.uk

Mobile site: http://m.spso.org.uk

14.1.7 You can obtain a copy of our complaints procedure by telephoning (0141 945 3282), e-mailing (enquiry@cadderhousing.co.uk) or by calling into our office.

15.0 AUDITING, MONITORING & REVIEW

15.1 Auditing

15.1.1 The Association has a commitment to ensure that the tenant selection and lettings procedure are carried out fairly, effectively and efficiently. In order to ensure that the allocation policy is being followed in a correct, fair and efficient manner we will undertake an audit of the allocations process. This self-assessment audit will measure our compliance with the provisions of our policies, procedures, legislative and good practice requirements. It will ensure that allocations have an audit trail that are clear and can account for why each allocation was made.

15.2 Monitoring & Reporting

- 15.2.1 The Acting Housing Manager will present information on the allocations process to the Management Committee on a quarterly basis, which will include: -
 - Demand information by street, size and type of property
 - Performance against Targets in the allocations process
 - ◆ Ethnic origin of applicants on the waiting lists and quarterly lettings
 - Number of applicants who identify they have a disability on the waiting lists and quarterly lettings
 - Gender of applicants on the waiting lists and quarterly lettings
 - Number of Void properties
 - ♦ Length of vacancies
 - Lost rent information
- 15.2.2 We will use this monitoring information to inform strategies to improve the allocations process, as well as ensure our policy continues to meet the needs of applicants. This information will also inform demand for all parts of our area and the review of our policy and procedures in the allocations and other relevant procedures, for example empty homes.
- 15.2.3 Results and outcomes of the allocations process will be monitored and reported to the Management Committee. This information will be used to inform the on-going review process. We will also provide data relating to our waiting lists to the Scottish Housing Regulator as part of our annual return on the social housing charter. The Charter sets the standards and outcomes that Cadder Housing Association should aim to achieve when performing its activities.

16.3 Review

- 16.3.1 This Policy is scheduled for review in December 2020 and will take account of:
 - Legislative, regulatory and good practice requirements
 - ♦ Association performance
 - ♦ the views of tenants staff and our members of the Management Committee.
- 16.3.2 As a result of the relevant areas within the Housing (Scotland) 2014
 Act, the Policy may be reviewed at an earlier date. This is dependent
 on a commencement date being put in place for the new measures
 within the Act.